Author’s response to reviews

Title: Lumbar Puncture as Possible Cause of Sudden Paradoxical Herniation in Patient with Previous Decompressive Craniectomy: Report of Two Cases

Authors:

Liang Shen (dliangshen@163.com)
Sheng Qiu (qius2001@126.com)
Zhongzhou Su (szzmdy@163.com)
Xudong Ma (xudongmahuzhou@162.com)
Renfu Yan (yanrenfu62@163.com)

Version: 4 Date: 07 Jul 2017

Author’s response to reviews:

Responses to comments 1 and 2 will be sent to BMCSeriesEditorial@biomedcentral.com.

Response to comment 3: Lumbar Puncture is common in neurosurgery. Previous Fig.1 and Fig.2 are removed from the manuscript.

Responses to comment 4 and 5: The checklist and the tracks are removed.